Author’s response to reviews

Title: Evaluating a Dementia Learning Community: exploratory study and research implications

Authors:

Rod Sheaff (r.sheaff@plymouth.ac.uk)

Ian Sherriff (ian.sherriff@plymouth.ac.uk)

Catherine Hennessy (chhennessy1@gmail.com)

Version: 1 Date: 14 Sep 2016

Author’s response to reviews:

The revisions are editorial requirement to prepare the manuscript for review. The changes made are:

1. Abbreviation 'PSDA' spelled out in full, in the abstract.

2. Informed consent of participants is now reported in Methods / Ethics section.